Author’s response to reviews

Title: Imaging study of coronal structural matching of the distal radius in normal adults

Authors:

Xin Zhang (2439299822@qq.com)
Feng Yuan (yuanfengtj@outlook.com)
Yong Yin (yinyong910428@126.com)
Jian Fan (qidongfanjian@sina.com)

Version: 3 Date: 19 Mar 2020

Author’s response to reviews:

March 19, 2020

Balkiran Sehmi Editors:
Journal of BMC Medical Imaging

Dear Editors:

Re: Revision of manuscript BMIM-D-19-00401R2
Thank you very much for your attention and the referee’s evaluation and comments on our paper “Imaging study of coronal structural matching of the distal radius in normal adults”. We have revised the manuscript according to your kind advices and detailed suggestions. Enclosed please find the responses to the referees. We sincerely hope this manuscript will be finally acceptable to be published on BMIM. Finally, all opinions on the format have met the requirements of your journal. Thank you very much for all your help and looking forward to hearing from you soon.

Yong Yin, Jian Fan
Yours sincerely,

Editor Comments:

1. Availability of data and materials
In the Availability of data and materials section please clearly state who should be contacted if someone wants to request the data
Response: Thank you very much for reading carefully and making pertinent suggestions. The relevant part has been amended. The datasets used and/or analysed during the current study are available from the corresponding author on reasonable request.
2. Ethics approval and consent to participate
In the Ethics approval and consent to participate section, please explicitly state whether any administrative permissions and/or licenses were acquired by your team to access the data used in your research.
Please remove the information in this section regarding where the CT images were taken from.
Response: The ethics committee of Tongji Hospital, School of Medicine, Tongji University approved the study.

3. Clean copy
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Additional files should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.
Response: Thank you very much for your suggestions and we have uploaded clean manuscript.
Finally, all opinions on the format have met the requirements of your journal.